Author's response to reviews

Title: Elevated IGFIR expression regulating VEGF and VEGF-C predicts lymph node metastasis in human colorectal cancer

Authors:

Chunhui Zhang (zhangchunhui2006@sohu.com)
Li Hao (lhao@yahoo.cn)
Liang Wang (wangl01@126.com)
Yichuan Xiao (xiao_yc@126.com)
Hailiang Ge (Geh@hotmail.com)
Zhenya Zhu (zyzhu@sohu.com)
Yunbao Luo (luoyb123@hotmail.com)
Yi Zhang (yizha@med.umich.edu)
Yanyun Zhang (yyzhang@sibs.ac.cn)

Version: 5 Date: 29 April 2010

Author's response to reviews:

Diana Marshall, PhD
Scientific Editor
BMC Cancer

RE: Manuscript. 2145346225302296

“Elevated IGFIR expression regulating VEGF and VEGF-C predicts lymph node metastasis in human colorectal cancer” by Chunhui Zhang, et al.

Dear Dr. Marshall:

Thank you very much for accepting our manuscript (2145346225302296) to publish in the journal of BMC Cancer. We made careful revisions to our manuscript according to the requirement point-by-point, please check, thank you very much!

Sincerely yours,

Yanyun Zhang, M.D. & Ph.D.
Shanghai Institute of Immunology
Institutes of Medical Sciences
Shanghai Jiao Tong University School of Medicine and
Key Laboratory of Stem Cell Biology
Institute of Health Sciences
Shanghai Institutes for Biological Sciences
Chinese Academy of Sciences &
Shanghai Jiao Tong University School of Medicine
227 South Chongqing Road, Shanghai 200025
China
Phone: +86-21-63846590 ext. 776769
Fax: +86-21-63852705
E-mail: yyzhang@sibs.ac.cn